Author's response to reviews

Title: A randomized cross over trial of tolerability and compliance of a micronutrient supplement with low iron separated from calcium vs high iron combined with calcium in pregnant women [ISRCTN56071145]

Authors:

Eric Ahn (eric.ahn@utoronto.ca)
Nicholas Pairaudeau (npairaudeau@hotmail.com)
Nicholas Pairaudeau Jr. (npairaudeau@hotmail.com)
Yves Cerat (yvescerat@videotron.ca)
Andre Fortier (andrej.fortier@videotron.ca)
Jean Deslauriers (jean.deslauriers@sympatico.ca)
Eric Paradis (ericparadis65@hotmail.ca)
Gideon Koren (gkoren@sickkids.ca)

Version: 7 Date: 17 March 2006

Author's response to reviews:

March 15, 2006

Dear Colleague:

In response to your message of Acceptance of March 9, 2006, enclosed please find a copy in which formatting changes have been made as requested, to comply with style.

Thank you for your help.

Sincerely,

Gideon Koren, MD, FABMT, FRCPC
Director, The Motherisk Program
Professor of Pediatrics, Pharmacology, Pharmacy, Medicine and Medical Genetics
The University of Toronto
Senior Scientist, The Research Institute
The Hospital for Sick Children
The Ivey Chair in Molecular Toxicology
University of Western Ontario
Canada